Author's response to reviews

Title: The Hypcholesterolemic Activity of Thai Glutinous Rice (Oryza sativa L. var. Niaw San-pah-tawng) Fermented by Monascus purpureus in High-Fat Cholesterol Diet Fed Rats

Authors:

Anurak Bunnoy (anurak.bunnoy@gmail.com)
Saisamorn Lumyong (scboi009@chaingmai.ac.th)
Siriwadee Chomdej (siriwadee@yahoo.com)
Supap Saenphet (bigc.cmu@gmail.com)
Kanokporn Saenphet (stit.lilo123@gmail.com)

Version: 3 Date: 19 September 2014

Author's response to reviews: see over
Dear editor,

Attached please find the revised manuscript (MS: 1037420545141680) entitled “The Hypocholesterolemic Activity of Thai Glutinous Rice (Oryza sativa L. var. Niaw San-pah-tawng) Fermented by Monascus purpureus in High-Fat Cholesterol Diet Fed Rats” by Kanokporn Saenphet, Anurak Bunnoy, Saisamorn Lumyong, and Supap Saenphet that has been made for the editorial assessment. The changes are as follows;

1. Insert line and page numbers into the manuscript.

2. Change the details of author’s names.

Since I bought the rice (rice seeds) from the local market, the identification of the rice was not possible. The details of plant identification, thus, are not provided.

I would very appreciate if you could kindly consider my manuscript for the peer review process. Thank you very much in advance for your kind consideration.

The name of suggested reviewers are as follows:

1. Assistant Professor Dr. Watee Kongbuntad
   Email: wateenaka@gmail.com
   Faculty of Science, Maejo University, Chiang Mai, Thailand

2. Professor Dr. Aranya Manosroi
   Email: a.manosroi@gmail.com
   Faculty of Pharmacy, Chiang Mai University, Chiang Mai, Thailand

Best regards

Assist. Prof. Dr. Kanokporn Saenphet

Department of Biology, Faculty of Science, Chiang Mai University, 239 Huay Kaew Road, Tumbol Suthep, Muang District, Chiang Mai, 50200, Thailand.
Phone: +66-86-605-0743, Fax: +66-53-892259